

Author's response to reviews

Title: Evaluation design of Urban Health Centres Europe (UHCE): preventive integrated health and social care for community-dwelling older persons in five European cities.

Authors:

Carmen Franse (c.franse@erasmusmc.nl)

Antonius Voorham (a.j.j.voorham@hr.nl)

Rob van Staveren (R.vanStaveren@zorgopnoord.nl)

Elin Koppelaar (e.koppelaar@hr.nl)

Rens Martijn (l.f.j.martijn@hr.nl)

Elisa Valia-Cotanda (Elisa.Valia@uv.es)

Tamara Alhambra-Borrás (tamara.alhambra@uv.es)

Tasos Rentoumis (development@vidavo.eu)

Lovorka Bilajac (lobilajac@gmail.com)

Vanja Vasiljev Marchesi (vvanja@medri.hr)

Tomislav Rukavina (tomislav.rukavina@medri.uniri.hr)

Arpana Verma (Arpana.Verma@manchester.ac.uk)

Greg Williams (greg.williams@manchester.ac.uk)

Gary Clough (gary.clough@manchester.ac.uk)

Jorge Garcés-Ferrer (jordi.garces@uv.es)

Francesco Mattace Raso (f.mattaceraso@erasmusmc.nl)

Hein Raat (h.raat@erasmusmc.nl)

Version: 1 Date: 04 Sep 2017

Author's response to reviews:

Dear Dr. Pasini,

We thank the editor for the opportunity to revise our manuscript and have adapted the manuscript according to the comments made. Below we address each comment

1. Title page

Please include the email of all authors in the title page.

We have included the email of all authors in the title page.

2. Abstract

Please change the Methods/Design header to Methods.

We have changed this header to 'Methods'.

3. Methods (main manuscript text)

Please change this section header to Methods/Design.

We have changed this header to 'Methods/Design'.

4. Figure

Please remove the figure from the main manuscript text and upload it as a separate file, making sure it does not contain the legend but just the image. The legend should be placed before the Reference, under the 'Figure legend' header.

We have removed the figure from the main manuscript text and uploaded it as a separate file. We have added the legend under the 'Figure legend' header.

5. Declarations

Please add this section header before the 'Acknowledgements' statement.

We have added the header 'Declarations'.

Thank you, yours sincerely,

Carmen Franse

Hein Raat